Author's response to reviews

Title: Effectiveness of second generation antipsychotics: A systematic review of randomized trials.

Authors:

Erik Johnsen (erij@ihelse.net)
Hugo A Jorgensen (hugo.jorgensen@psyk.uib.no)

Version: 3 Date: 22 April 2008

Author's response to reviews: see over
Dear Sirs,

We have made the following revisions as requested, and hope they are to your satisfaction:

1. Statements on funding of our systematic review are in the Acknowledgements section. The other statements on funding relate to the included papers in this review, not the review itself. We have included these in the Methods and Results sections as the matter of funding is one of the outcome measures of our review. Should these also be in the Acknowledgements section?
2. The figure legend has been appropriately titled.
3. The figure is cropped as closely as possible.

Yours sincerely,

Erik Johnsen, M.D.     Hugo A. Jørgensen. M.D., Ph.D.